Author’s response to reviews

Title: Community-based psychosocial interventions for people with schizophrenia in low and middle-income countries: systematic review and meta-analysis

Authors:

Laura Asher (laura.asher@lshtm.ac.uk)

Vikram Patel (vikram_patel@hms.harvard.edu)

Mary De Silva (M.Desilva@wellcome.ac.uk)

Version: 3 Date: 16 Oct 2017

Author’s response to reviews:

I have made the minor changes to the section titles as requested. I have added a short conclusion to the manuscript (which is the same as the conclusion section in the abstract).